

# Biallelic variants in TRAPPC10 cause a microcephalic TRAPPopathy disorder in humans and mice

Lettie E Rawlins, Hashem Almousa, Shazia Khan, Stephan C Collins, Miroslav P Milev, Joseph Leslie, Djenann Saint-Dic, Valeed Khan, Ana Maria Hincapie, Jacob O Day, et al.

### ▶ To cite this version:

Lettie E Rawlins, Hashem Almousa, Shazia Khan, Stephan C Collins, Miroslav P Milev, et al.. Biallelic variants in TRAPPC10 cause a microcephalic TRAPPopathy disorder in humans and mice. PLoS Genetics, 2022, 18 (3), pp.e1010114. 10.1371/journal.pgen.1010114. inserm-03944752

## HAL Id: inserm-03944752 https://inserm.hal.science/inserm-03944752v1

Submitted on 18 Jan 2023  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



## G OPEN ACCESS

**Citation:** Rawlins LE, Almousa H, Khan S, Collins SC, Milev MP, Leslie J, et al. (2022) Biallelic variants in *TRAPPC10* cause a microcephalic TRAPPopathy disorder in humans and mice. PLoS Genet 18(3): e1010114. https://doi.org/10.1371/journal.pgen.1010114

**Editor:** Gregory S. Barsh, HudsonAlpha Institute for Biotechnology, UNITED STATES

Received: September 26, 2021

Accepted: February 20, 2022

Published: March 17, 2022

**Copyright:** © 2022 Rawlins et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The authors are grateful for funding support provided by Wellcome Trust (209083/Z/ 17/Z to ELB) https://wellcome.org/grant-funding, the Newlife Foundation for Disabled Children (AHC, LER and ELB) https://newlifecharity.co.uk/, Mitacs Globalink research award (IT14910 to LER) https:// www.mitacs.ca/en/programs/globalink and Higher Education Commission (HEC), Pakistan (IRSIP I-8/ **RESEARCH ARTICLE** 

# Biallelic variants in *TRAPPC10* cause a microcephalic TRAPPopathy disorder in humans and mice

Lettie E. Rawlins <sup>1,2°</sup>, Hashem Almousa <sup>3°</sup>, Shazia Khan<sup>1,4°</sup>, Stephan C. Collins <sup>5,6°</sup>, Miroslav P. Milev<sup>3</sup>, Joseph Leslie <sup>1</sup>, Djenann Saint-Dic <sup>3</sup>, Valeed Khan <sup>7</sup>, Ana Maria Hincapie<sup>3</sup>, Jacob O. Day <sup>1,8</sup>, Lucy McGavin<sup>9</sup>, Christine Rowley<sup>10</sup>, Gaurav V. Harlalka <sup>1,11</sup>, Valerie E. Vancollie <sup>10</sup>, Wasim Ahmad<sup>12</sup>, Christopher J. Lelliott <sup>10</sup>, Asma Gul<sup>4</sup>, Binnaz Yalcin <sup>5,6°</sup>, Andrew H. Crosby <sup>1°</sup>, Michael Sacher <sup>3,13°</sup>, Emma L. Baple <sup>1,2°</sup>\*

1 RILD Wellcome Wolfson Medical Research Centre, RD&E (Wonford) NHS Foundation Trust, University of Exeter Medical School, Exeter, United Kingdom, 2 Peninsula Clinical Genetics Service, Royal Devon & Exeter Hospital (Heavitree), Exeter, United Kingdom, 3 Department of Biology, Concordia University, Montreal, Quebec, Canada, 4 Department of Biological Sciences, International Islamic University, Islamabad, Pakistan, 5 Institute of Genetics and Molecular and Cellular Biology, Inserm, Illkirch, France, 6 Inserm, University of Bourgogne Franche-Comté, Dijon, France, 7 Department of Molecular Diagnostics, Rehman Medical Institute, Peshawar, Pakistan, 8 Faculty of Health, University of Plymouth, Plymouth, United Kingdom, 9 University Hospitals Plymouth NHS Trust, Plymouth, United Kingdom, 10 Wellcome Sanger Institute, Hinxton, Cambridge, United Kingdom, 11 Department of Pharmacology, Rajarshi Shahu College of Pharmacy, Malvihir, Buldana, India, 12 Department of Biochemistry, Faculty of Biological Sciences, Quaidi-Azam University, Islamabad, Pakistan, 13 Department of Anatomy and Cell Biology, McGill University, Montreal, Quebec, Canada

These authors contributed equally to this work.

\* E.Baple@exeter.ac.uk

## Abstract

The highly evolutionarily conserved transport protein particle (TRAPP) complexes (TRAPP II and III) perform fundamental roles in subcellular trafficking pathways. Here we identified biallelic variants in TRAPPC10, a component of the TRAPP II complex, in individuals with a severe microcephalic neurodevelopmental disorder. Molecular studies revealed a weakened interaction between mutant TRAPPC10 and its putative adaptor protein TRAPPC2L. Studies of patient lymphoblastoid cells revealed an absence of TRAPPC10 alongside a concomitant absence of TRAPPC9, another key TRAPP II complex component associated with a clinically overlapping neurodevelopmental disorder. The TRAPPC9/10 reduction phenotype was recapitulated in TRAPPC10<sup>-/-</sup> knockout cells, which also displayed a membrane trafficking defect. Notably, both the reduction in TRAPPC9 levels and the trafficking defect in these cells could be rescued by wild type but not mutant TRAPPC10 gene constructs. Moreover, studies of Trappc10<sup>-/-</sup> knockout mice revealed neuroanatomical brain defects and microcephaly, paralleling findings seen in the human condition as well as in a  $Trappc9^{\gamma}$ mouse model. Together these studies confirm autosomal recessive TRAPPC10 variants as a cause of human disease and define TRAPP-mediated pathomolecular outcomes of importance to TRAPPC9 and TRAPPC10 mediated neurodevelopmental disorders in humans and mice.

HEC/HRD/2017/6914 to SK) https://www.hec.gov. pk/english/pages/home.aspx. Functional studies were supported by the Canadian Institutes of Health Research (to MS) https://cihr-irsc.gc.ca/e/ 193.html and the Natural Sciences and Engineering Research Council of Canada (to MS) https://www. nserc-crsng.gc.ca/index\_eng.asp. The mouse neuroanatomical studies were supported by The French National Research Agency (ANR-18-CE12-0009 JCJC to BY) https://anr.fr/en/anrs-role-inresearch/missions/. This work was also supported by Wellcome Trust grant WT098051 (to BY) https://wellcome.org/grant-funding. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Microcephalic neurodevelopmental disorders are a group of conditions that are often inherited in families, involving small head size and abnormal brain development and function. This often results in delayed development of an affected child, affecting their movement, language and/or non-verbal communication and learning, as well as seizures and neuropsychiatric problems. A group of proteins called the transport protein particles (TRAPPs) are important for the transport of cargos inside cells. Alterations within a number of the TRAPP proteins have previously been associated with human inherited diseases called the 'TRAPPopathies', which involve neurodevelopmental and skeletal abnormalities. Here we show that TRAPPC10 gene alterations cause a new TRAPPopathy microcephalic neurodevelopmental disorder, and we provide a detailed clinical description of the condition termed 'TRAPPC10-related disorder'. Our studies in mice lacking the TRAPPC10 gene identified similar features to those of affected humans, including small brain size and skeletal abnormalities. Our molecular studies showed that an affected individual with an alteration in the TRAPPC10 gene has no functional TRAPPC10 protein in their cells, which in turn causes a reduction in levels of another important TRAPP molecule, TRAPPC9. Cells lacking TRAPPC10 also display abnormalities in cellular transport processes. Together our data confirm alterations in TRAPPC10 as a cause of a microcephalic neurodevelopmental disorder in both humans and mice.

#### Introduction

The transport protein particles (TRAPPs) comprise highly evolutionarily conserved multiprotein complexes, originally identified in yeast. Three yeast TRAPP complexes have been described (TRAPP I, II and III) which play important roles in secretory and endocytic subcellular trafficking pathways [1,2]. Humans possess only two complexes (TRAPP II and III), both sharing the same core proteins alongside additional complex-specific components, some not identified in yeast. Previous studies have identified diverse cellular roles for the TRAPP II complex including Golgi, COP (coat protein)-I vesicular, lipid droplet, centrosomal and ciliary functions [3-5]. The TRAPP II complex contains two complex-specific subunits; TRAPPC9 and TRAPPC10 [6]. An additional TRAPP II complex-associated protein has recently been described (TRAPPC14/C7orf43) [7]. However, TRAPPC14 does not appear to be required for TRAPP II complex-associated guanine nucleotide exchange factor activity [8,9] and lack of the molecule does not affect TRAPP II complex assembly or stability [7]. Thus, further studies are required to clarify the role of TRAPPC14 in TRAPP II. TRAPPC9 has been implicated in COP II vesicle trafficking [10] and NF- $\kappa$ B signalling pathway activation [11], with TRAPPC9 depletion associated with increased lipid droplet formation [4]. TRAPPC14 plays a role in Rabin 8 binding to the TRAPP II complex and its tethering of preciliary vesicles to the mother centriole during ciliogenesis [7]. No detailed studies of TRAPPC10 function have been conducted, although it has been shown to interact with TRAPPC2L and may have roles in guanine nucleotide exchange factor (GEF) activity regulation [12].

Ten of the 15 TRAPP complex proteins have been associated with human inherited diseases, collectively termed "TRAPPopathies" [6]. This includes TRAPPC2, the only subunit associated with skeletal dysplasia (spondyloepiphyseal dysplasia tarda (SEDT); OMIM #313400) [13], and nine other TRAPP subunits (TRAPPC2L, TRAPPC4, TRAPPC6A, TRAPPC6B, TRAPPC9, TRAPPC10, TRAPPC11, TRAPPC12 and TRAPPC14) associated with neurodevelopmental disorders [12,14–24]. While diverse, the clinical features of these disorders are overlapping and include neurodevelopmental delay, intellectual disability, developmental regression, primary and postnatal-onset microcephaly, epilepsy, hypotonia, craniofacial dysmorphism and structural brain abnormalities including thinning of the corpus callosum and reduced white matter volume (S1 Table).

Of the TRAPP II complex specific subunits, only TRAPPC9 has been robustly associated with human disease, an autosomal recessive neurodevelopmental disorder (OMIM #613192) characterized by postnatal-onset microcephaly with reduced white matter volume and corpus callosum thinning, intellectual disability, dysmorphic features, hypotonia, epilepsy, and raised body mass index [18–20]. TRAPP II plays important roles in Golgi membrane trafficking. "Golgipathy" neurodevelopmental disorders including the TRAPPopathies comprise an expansive group of conditions associated with variants in Rab GTPases, conserved oligomeric Golgi (COG) complex, and coat protein (COP) complex proteins among others [25,26] and are characterized by primary or postnatal-onset microcephaly, intellectual disability, seizures and white matter brain abnormalities.

Here, we present comprehensive genetic, clinical, functional and mouse data to define biallelic *TRAPPC10* gene variants as a cause of a microcephalic neurodevelopmental disorder, providing important insight into the TRAPPopathy family of disorders.

#### Results

# Biallelic *TRAPPC10* variants are associated with a microcephalic neurodevelopmental disorder

Table 1 summarizes the clinical features of affected individuals in this study. Family 1 comprises a large interlinking multi-nuclear Pakistani family, with eight individuals (4.1-18.1 years) affected by a severe microcephalic neurodevelopmental disorder (Fig 1A, Family 1). All but one individual presented with microcephaly (>3 standard deviations (SDS) below mean), and all displayed short stature (-1.78 to -5.09 SDS). All eight were born at full term, prenatal histories were unremarkable, although antenatal ultrasound scans were not performed. The parents did not report that any affected infant's head was noticeably small (no weight or occipitofrontal circumference (OFC) measurements were recorded). The earliest OFC measurement documented was at 2.5 years (1-IV:4), and at age 4.1 years his microcephaly had not worsened. Affected individuals displayed mild craniofacial dysmorphism (S1A-S1C Fig). Although standardized intelligence quotient (IQ) testing was not possible due to cultural and language barriers, all affected individuals were assessed as severely intellectually impaired (DSM-5 criteria). Developmental trajectories were similar in all, as was the degree of impairment across all developmental domains. Hypotonia in infancy was universal and walking was delayed to between 2.5-4 years. All affected individuals have speech impairment, five have only a few words and three are non-verbal. Behavioural abnormalities including autistic features and aggressive episodes were a consistent feature. A history of seizures from early infancy was reported in four individuals, one has ongoing generalised tonic-clonic seizures (1-IV:2) and the remaining three (1-IV:3, 1-IV:7, 1-IV:13) are seizure free on medication. Neuroimaging was only performed for one case (1-IV:2) due to the remote family location and as sedation/general anaesthesia was deemed necessary. This revealed microcephaly with corpus callosum thinning and no other abnormalities (Fig 1B and 1C). All had normal systemic examination findings and no evidence of gross skeletal disproportion (S1A Fig). It was only possible to obtain skeletal radiographs for a single individual (1-IV:12) aged 9 years, these were unremarkable but unfortunately spinal imaging was incomplete.

FAMILY				FAMI	1 7 1				FAMI	LY 2
Pedigree reference	IV:2	IV:3	IV:4	IV:7	IV:8	IV:10	IV:12	IV:13	IV:1	IV:2
Genotype	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(G1131Vfs*19)/ p.(G1131Vfs*19)	p.(P929L)/ p.(P929L)	p.(P929L)/ p.(P929L)
Gender	W	M	W	H	Н	Н	W	M	W	M
Age at evaluation (years)	18.1	13.8	4.1	14.1	15.6	5.5	0.6	13.7	25	22
GROWTH										
Birth weight kg (SDS)	NK	NK	2.5 (-2.25)	NK	NK	NK	NK	NK	NK	NK
OFC cm (SDS)	55 (-1.32)	50 (-3.49)	47 (-3.59)	46 (-6.64)	49 (-4.56)	43 (-7.46)	43 (-7.05)	47 (-5.28)	50 (-4.24)	53 (-2.49)
Height cm (SDS)	164 (-1.87)	146 (-1.78)	94 (-2.23)	127 (-5.09)	142 (-3.41)	99 (-2.78)	119 (-2.48)	125 (-4.26)	150 (-3.92)	140 (-5.29)
Weight kg (SDS)	52 (-1.96)	40 (-0.99)	18 (0.6)	25 (-4.9)	45 (-1.39)	16 (-1.44)	20 (-2.77)	28 (-3.16)	45 (-3.9)	38 (-4.87)
BMI	19.3	18.7	20.4	15.5	22.3	16.3	14.1	17.9	20	19
DEVELOPMENT										
Intellectual disability	Severe	Severe	Severe	Severe	Severe	Severe	Severe	Severe	Severe	Severe
Global developmental delay	>	>	>	>	>	>	>	>	>	>
Speech impairment	>	>	>	>	>	>	>	>	>	>
4	<10 words	Non-verbal	<10 words	Non-verbal	<10 words	Non-verbal	<10 words	<10 words		
Walked (years)	4	4	Standing	3.5	2.5	4	4	Э	NK	NK
Hearing	No concerns	Otitis media	No concerns	No concerns	No concerns	No concerns	No concerns	No concerns	No concerns	No concerns
Vision	No concerns	No concerns	No concerns	No concerns	No concerns	No concerns	Strabismus	No concerns	Strabismus	Strabismus
NEUROLOGY										
Childhood hypotonia	>	`	>	>	>	>	>	>	NK	NK
Seizures	>	>		>				>		
Gait abnormalities		Waddling gait					Waddling gait	Waddling gait	NK	NK
Behavioural abnormalities	>	`	>	>	>	>	>	>	>	>
Abbreviations: M	, male; F, female; N	VK, not known; SDS	standard deviatior	1 score; OFC, occipi	itofrontal circumfer	ence; G, glycine; V	, valine; fs, frameshi	ft; (∕), indicates pı	esence of a fe	ature in an
affected individu	ıl; (-), indicates abs	sence of a feature in	an affected individu	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	and OFC Z-scores v	vere calculated usin	g a Microsoft Excel	add-in to access gr	owth reference	es based on
the LMS method <sup>6</sup>	using UK 1990 rei	ference population <sup>b</sup>								
<sup>a</sup> Pan H, Cole TJ.	LMS growth, a Mi	crosoft Excel add-ir	to access growth re	eferences based on t	he LMS method. V	ersion 2.77, http://v	www.healthforallchil	ldren.co.uk/;2012		
<sup>b</sup> Cole TJ, Freems	m JV, Preece MA:	British 1990 growth	reference centiles fo	or weight, height, bo	ody mass index and	head circumferend	e fitted by maximur	n penalized likelih	ood. Stat Med	1998, 17

https://doi.org/10.1371/journal.pgen.1010114.t001

(4):407-4.



**Fig 1. Biallelic** *TRAPPC10* variants identified in two families with individuals affected by a microcephalic neurodevelopmental disorder. (A) Family 1 Extended pedigree comprising three interlinking families and a total of eight affected individuals all of whom are homozygous for the *TRAPPC10* c.3392delG; p.(Gly1131Valfs\*19) variant ('+'), with co-segregation confirmed in other family members ('-' indicates the wild type allele). **Inset:** Dideoxy sequence chromatogram of an affected individual homozygous for the *TRAPPC10* c.3392delG; p.(Gly1131Valfs\*19) variant. **Family 2** Previously reported family (MR107) [27] showing co-segregation of the *TRAPPC10* c.2786C>T; p.(Pro929Leu) variant ('+') in a four generation pedigree ('-' indicates the wild type allele). **Inset:** Dideoxy sequence chromatogram of an affected individual homozygous for the *TRAPPC10* c.2786C>T; p.(Pro929Leu) variant. **(B, C)** MRI neuroimaging of individual IV:2 from family 1 demonstrates microcephaly and profound thinning of the corpus callosum with no other abnormalities. (**D**) Genome-wide SNP mapping in seven affected individuals (1-IV:2–4, 1-IV:7–8 and 1-IV:12–13) identified a single (3.5Mb) region of shared homozygosity, containing 91 genes including *TRAPPC10*. (**E**) Schematic diagram of the TRAPPC10 protein identifying *TRAPPC10* sequence variants and TRAPP domain. (**F**) Multi-species alignment showing conservation of the molecular region encompassing the p.Gly1131Valfs\*19 and p.Pro929Leu variants.

https://doi.org/10.1371/journal.pgen.1010114.g001

Assuming homozygosity for a founder variant was responsible, we undertook SNP genotyping (1-IV:2–4, 1-IV:7–8, 1-IV:12–13) in parallel with exome sequencing (1-IV:2) to identify the cause of disease. SNP mapping identified a single notable (>1Mb) homozygous region common to all affected individuals (3.5Mb, rs2839641-rs7278087; chr21:g.43,179,611– 46,678,912 [hg38]) as the likely disease locus (Fig 1D). Concomitant exome sequencing data analysis to identify candidate homozygous and compound heterozygous rare (<0.01 frequency) variants predicted to have a functional impact, identified a single candidate variant within this region which could not be excluded. This frameshift variant (chr21:g.44102823del; NM\_003274.4:c.3392del; p.(Gly1131Valfs\*19)[hg38]; Fig 1A) in the penultimate (22/23) exon of *TRAPPC10* located in the chromosome 21q22.12 region, is present within the C-terminal TRAPP domain (Fig 1E) and is predicted to result in nonsense mediated mRNA decay of the mutant transcript (Fig 1F). The variant was confirmed in the lymphocyte RNA of an affected individual (S2 Fig), is not listed in publicly available genomic databases (including gnomAD v2.1.1/v3.1.1) and co-segregates appropriately (Fig 1A).

This finding enabled us to revisit the genetic data and more comprehensively phenotype two siblings in an unrelated Pakistani family (Fig 1A, Family 2) included in a larger study to identify novel candidate neurodevelopmental disease genes ([27] Family MR107). Both male siblings (2-IV:1, 22 years and 2-IV:2, 25 years), were described to have "severe intellectual disability, aggressive behaviour and poor speech". A homozygous exon 18 *TRAPPC10* variant (chr21:g.44089849C>T; NM\_003274.4:c.2786C>T; p.(Pro929Leu) [hg38]) (Fig 1A) was identified as a potential cause of disease [27]. This variant, which co-segregated appropriately, is absent from gnomAD v2.1.1/v3.1.1. Multiple species alignment confirms stringent conservation of Pro929 across vertebrates (Fig 1F). Additional clinical features noted and not previously reported, included microcephaly/borderline microcephaly (-2.49 SDS, -4.2 SDS), short stature (-3.86, - 5.29 SDS) and mild dysmorphism, all consistent with Family 1.

#### TRAPPC10 variants exhibit reduced interaction with TRAPPC2L, destabilize the TRAPP II complex and result in an anterograde trafficking defect

We next sought to determine the molecular consequences of each TRAPPC10 genetic variant. As we previously determined that TRAPPC10 interacts with TRAPPC2L in the TRAPP II complex [12], we used yeast two-hybrid assays to investigate how TRAPPC10 alteration affects TRAPPC2L interactions. When TRAPPC10 was in the bait (pGBKT7) or prey (pGADT7) vector, TRAPPC2L interaction was observed (Fig 2A and 2B). However, both p.Gly1131Valfs\*19 and p.Pro929Leu variants showed reduced TRAPPC2L interaction (Fig 2A and 2B), though the proteins were expressed in the yeast cells (not shown). This was confirmed quantitatively by measuring  $\beta$ -galactosidase activity (Fig 2C), one of the yeast two-hybrid system markers. Given this, we generated a patient lymphoblastoid cell line (LCL) (1-IV:2; TRAPPC10 p. (Gly1131Valfs\*19)) to investigate TRAPP complex outcomes. We examined core TRAPP protein (TRAPPC2, TRAPPC2L, TRAPPC3) levels, as well as TRAPP III-specific (TRAPPC8, TRAPPC12), and TRAPP II-specific (TRAPPC10, TRAPPC9) levels. TRAPPC2L, TRAPPC2 and TRAPPC3 levels were slightly reduced in the p.(Gly1131Valfs\*19) lysate compared to control (Fig 2D). As expected we did not detect TRAPPC10 as the commercially available TRAPPC10 antibody is directed to the molecular region downstream (C-terminal) to the p. (Gly1131Valfs\*19) alteration. Intriguingly we also noted an absence of full length TRAPPC9, although in contrast, TRAPP III-specific proteins levels were not affected. We then fractionated p.(Gly1131Valfs\*19) and control LCL lysates on a size exclusion column. While TRAPPC10 and TRAPPC9 fractionated as expected in controls, they were absent from p.



**Fig 2. TRAPP II complex integrity is affected by TRAPPC10 variants and in TRAPPC10**<sup>-/-</sup> **cells.** Yeast cells transformed with pGBKT7 or pGBKT7 expressing either wild type TRAPPC10, the p.Gly1131Valfs\*19 variant or the p.Pro929Leu variant (**A**), or pGADT7 or pGADT7

expressing either wild type TRAPPC10, the p.Gly1131Valfs<sup>\*</sup>19 variant or the p.Pro929Leu variant (**B**) were mated with yeast expressing either pGBKT7-TRAPPC2L (**A**) or pGADT7-TRAPPC2L (**B**). Serial dilutions of the diploid cells were grown on SC medium lacking leucine and tryptophan (DDO) or SC medium lacking leucine, tryptophan, histidine and adenine (QDO). (**C**) Diploid cells from (**A**) and (**B**) were quantitatively tested for  $\beta$ -galactosidase activity. Units of activity were calculated according to the following formula: (OD<sub>420</sub> x 1000)/(OD<sub>600</sub> x time in hours). Note that western analysis revealed that all forms of TRAPPC10 and the two variants were expressed in the yeast cells. (**D**) Lymphoblastoid cells from control or an individual homozygous for the TRAPPC10 p.(Gly1131Valfs<sup>\*</sup>19) variant were lysed and probed for the TRAPP Proteins indicated and for tubulin as a representative loading control. The lysates were fractionated on a Superose 6 size exclusion column. Fractions of 0.5ml were collected. Neighbouring fractions were pooled and fractionated by SDS-PAGE and probed for the ubulin as a representative loading control. The Issates were fractions of 0.5ml were collected. Neighbouring fractions were pooled and probed for the TRAPP proteins indicated and for tubulin as a representative loading control. The lysates were fractionated on a Superose 6 size exclusion column. Fractions of 0.5ml were tractionated on a Superose 6 size exclusion column. Fractions of 0.5ml were proteins indicated and for tubulin as a representative loading control. The lysates were fractionated and for tubulin as a representative loading control. The lysates were fractionated on a Superose 6 size exclusion column. Fractions of 0.5ml were collected. Neighbouring fractions were pooled and fractionated by SDS-PAGE and probed for the indicated TRAPP proteins. The fractionation of molecular size standards are indicated above the top-most panel of the size exclusion portion of (**D**) and (**E**).

https://doi.org/10.1371/journal.pgen.1010114.g002

(Gly1131Valfs\*19) lysate (Fig 2D) indicating an absence of TRAPP II (peak in fraction 28 of control cells). There was no change in core protein (TRAPPC2, TRAPPC2L, TRAPPC3) fractionation consistent with their presence in TRAPP III, although the portion that co-fractionates with TRAPP II was reduced. Similarly, no change in TRAPP III proteins (TRAPPC8, TRAPPC12) fractionation was seen, indicating that this complex was not affected in p. (Gly1131Valfs\*19) cells.

We then examined the consequences of TRAPPC10 loss on membrane trafficking. Since we were unable to perform trafficking assays in smaller LCLs, we generated HEK293 TRAPPC10 knockout cell lines using CRISPR/Cas9. Three different TRAPPC10 knockout lines were generated (S3 Fig), two were examined and behaved identically (TRAPPC10<sup>-/-</sup> KO1/KO7); as such we present data from one line (KO7, hereafter referred to as TRAPPC10<sup>-/-</sup>). We first examined TRAPP protein levels in TRAPPC10<sup>-/-</sup> cells. Consistent with observations in LCLs, we noted a slight reduction in core TRAPP proteins, no effect on TRAPP III-specific proteins (TRAPPC8, TRAPPC12), and the absence of TRAPPC9, consistent with size exclusion chromatography results (Fig 2E). We then performed the VSVG-GFP ts045 trafficking assay [28] in TRAPPC10<sup>-/-</sup> cells. In this assay, the marker protein VSVG-GFP ts045 is retained in the ER at elevated temperature, but synchronously released at 32°C. The fluorescently-tagged marker protein accumulated in the Golgi region in controls, with peak signal at ~25 minutes (Fig 3A, black curve). From that point onward the signal was cleared from the Golgi as the protein travelled to the plasma membrane. In contrast, the arrival into, trafficking through and release from the Golgi of the marker protein in TRAPPC10<sup>-/-</sup> cells was delayed compared to controls (Fig 3A, green curve). Importantly, the trafficking delay was shown to be TRAPPC10-dependent as cell transfection with wild type (WT) TRAPPC10 resulted in a trafficking defect rescue (Fig 3A, blue curve). Although both p.Gly1131Valfs\*19 and p.Pro929Leu variants were able to partially rescue these defects, neither rescued anterograde trafficking to the same extent as WT protein (Fig 3A, red/grey curves). Taken together, these results demonstrate that the TRAPPC10 variants cause partial functional defects, and that the absence of TRAPPC10 results in a concomitant absence of TRAPPC9.

# Expression of TRAPPC10 is important for the detection of a cellular pool of TRAPPC9

As no antibody was available to determine whether mutant TRAPPC10 p.(Gly1131Valfs\*19) polypeptide may be expressed in affected individuals, we generated WT and both (p. Gly1131Valfs\*19, p.Pro929Leu) constructs to investigate protein stability. The constructs were transfected into HEK293 cells and lysates prepared daily for four days. As shown in Fig 3B, WT protein levels on day 4 dropped to ~70% of that on day 1. In contrast, levels of both altered TRAPPC10 proteins (day 4) dropped to 25–50% (day 1), with the drop in the p.



Fig 3. Cells devoid of TRAPPC10 display a membrane trafficking defect and lack detectable levels of TRAPPC9. (A) Wild type HEK293 cells or TRAPPC10<sup>-/-</sup> cells either not transfected or transfected with FLAG-tagged wild type TRAPPC10 or one of the TRAPPC10 variants indicated were infected with VSVG-GFP ts045

4 hours after transfection. After an overnight incubation at 40°C, the cells were shifted to 32°C and imaged every minute. The fluorescence intensity in the Golgi was quantified and plotted versus time as described in the Materials and Methods section. The inset shows a western blot for the transfected proteins probed with anti-FLAG antibody. Representative images are shown in S7 Fig. (**B**) RFP-tagged versions of either wild type TRAPPC10, the p.Gly1131Valfs\*19 variant or the p. Pro929Leu variant were transfected into HeLa cells and lysates prepared every day for 4 days. The lysates were fractionated by SDS-PAGE and probed with anti-mCherry antibody to reveal the RFP-tagged construct. Though the trends were consistent, statistical significance was seen only at day 4. The half-life for each overexpressed protein was found to be approximately 5.4 days, 4.4 days and 3.1 days for the wild type, p.Gly1131Valfs\*19 and p.Pro929Leu variants, respectively. (**C**) RFP-tagged TRAPPC10 and the p.Gly1131Valfs\*19 and p.Pro929Leu variants were transfected into HEK293 cells. The cells were either untreated (-) or treated (+) with 50 µm MG132 for 24 hours on the third day post-transfected or transfected or transfected by SDS-PAGE and probed with anti-mCherry and tubulin as a loading control. (**D**) TRAPPC10<sup>-/-</sup> cells were either untransfected or transfected with FLAG constructs to verify expression, TRAPPC9 and tubulin as a loading control. Parental HEK293 cells were also probed to assess the level of TRAPPC10 in the presence of TRAPPC10. (**E**) Lysates were prepared from lymphoblastoid cells from either control, an individual homozygous for the TRAPPC10 in the presence of TRAPPC10. (**E**) Lysates were prepared from TRAPPC9 p.(Arg475\*) variant and subjected to western analysis to reveal the proteins indicated.

https://doi.org/10.1371/journal.pgen.1010114.g003

Gly1131Valfs\*19 protein being most dramatic. The decrease in expression of WT and both p. Gly1131Valfs\*19 and p.Pro929Leu TRAPPC10 proteins was blocked by inclusion of proteasome inhibitor MG132 (Fig 3C). These results suggest that the two mutant proteins are subject to enhanced degradation compared to WT protein.

The absence of TRAPPC9 in TRAPPC10 p.(Gly1131Valfs\*19) LCLs prompted us to closely examine the inter-relationship between TRAPPC9 and TRAPPC10. We first investigated whether TRAPPC10 expression in TRAPPC10<sup>-/-</sup> cells would restore TRAPPC9 levels. Upon transfection of WT TRAPPC10 into TRAPPC10<sup>-/-</sup> cells, the appearance of TRAPPC9 was noted, although not to parental HEK293 cell levels (Fig 3D). Overexpression of the p.Pro929Leu variant, and to a much lesser degree p.Gly1131Valfs\*19, restored some TRAPPC9 protein although again not to the same extent as WT TRAPPC10.

We next investigated whether a reciprocal absence of TRAPPC10 occurred in cells devoid of TRAPPC9. We probed LCL lysates from three individuals with the p.(Arg475\*) TRAPPC9 variant [19]. As expected TRAPPC9 was not detected in these cells (Fig 3E), nor was it detected in TRAPPC10 p.(Gly1131Valfs\*19) cells (as shown above). Importantly, while TRAPPC10 levels were significantly reduced in p.(Arg475\*) TRAPPC9 lysates, the protein was nevertheless present. Collectively, these results suggest that the detection of a cellular pool of TRAPPC9 is strongly dependent upon the expression of full length TRAPPC10.

# *Trappc10<sup>-/-</sup>* mice have similar neurodevelopmental deficits to patients with biallelic *TRAPPC10* variants

We then evaluated tissue sections and datasets from a *Trappc10<sup>tm1b(EUCOMM)Wtsi*</sub> mouse model, with a focus on the phenotypical components of the human disorder in particular neuroanatomical findings. At weaning age, mouse survival was evaluated from successfully genotyped mice originating from multiple litters derived from a heterozygous-by-heterozy-gous breeding scheme. We obtained the expected number of WT, heterozygous and homozy-gous mice. Male and female mice were weighed the same day each week from 4 until 16 weeks of age (S4 Fig).</sup>

Using a recently developed robust approach to assess 63 brain parameters across 23 brain regions [29], we analyzed neuroanatomical defects in adult  $Trappc10^{-/-}$  mice blinded for genotype. To minimize environmental and genetic variation, male mice aged 16 weeks were used. In the homozygous mutant mice, many neuroanatomical parameters were reduced in size when compared to WTs (Fig 4A and 4B). The total brain area parameter was significantly reduced (14%, P = 0.0003), concomitantly with smaller white matter structure size including the genu of the corpus callosum (-25%, P = 0.026), hippocampal commissure (-38%, P = 0.013), hippocampal fimbria (-17%, P = 0.0049), anterior commissure (-28%, P = 3.8E10<sup>-8</sup>)



**Fig 4. Mouse studies implicate TRAPPC10 in the formation of white matter structures. (A)** Histograms for three homozygous *Trappc10<sup>-/-</sup>* mice showing variation (decreased-minus scale or increased-positive scale) in areas and lengths expressed as percentage of 498 WTs together with a color map indicating the significance level (white indicates not significant). The list of measurements and corresponding numbers are shown below the histograms. (**B**) Schematic representation of a section at Bregma +0.98 mm and Bregma -1.34 mm. Colored regions indicate the presence of at least one significant parameter within the brain region at the 0.05

level. (C-D) Nissl-stained sagittal brain sections from  $Trappc10^{im1b/m1b}$  mice, showing the anterior commissure (C) and the total brain area and internal capsule size reduction in mutants (D). The corresponding scale is shown in each panel. (E) Box plots of combined grey and white matter structures expressed in cm<sup>2</sup> using a set of 37 local WTs (same genetic background, housing conditions, age, sex, experimental and necropsy dates) compared to three  $Trappc10^{im1b/tm1b}$  mice. (F-G) Box plots with raw data points detailing the cellular and myelination characterisation of the anterior commissure (F) and internal capsule (G) using a set of 15 local WTs. Statistical analyses were performed with GraphPad Prism 8.0.2, using two-tailed Student's *t*-tests of equal variances. \*p<0.05 \*\*p<0.01 \*\*\*p<0.001.

https://doi.org/10.1371/journal.pgen.1010114.g004

and the internal capsule (-23%,  $P = 1.1E10^{-6}$ ) (Fig 4C and 4D). When combining neuroanatomical parameters from either grey or white matter structures, only white matter structures were reduced in size (-20%,  $P = 8E10^{-6}$ ) (Fig 4E). Interestingly, the size reductions of the anterior commissure and internal capsule were correlated with loss of myelination whilst oligodendrocyte cell population count and density were unaffected (Fig 4F and 4G). Together these findings suggest that both mouse and human microcephaly stems from defects in white matter structures and myelin biogenesis.

Given the short stature exhibited by patients with *TRAPPC10*-related disorder we also evaluated *Trappc10<sup>-/-</sup>* mice growth parameters (S1 Text). While long bone length was marginally reduced in both sexes, it only reached significance for female *Humerus* (S4A Fig), although body length overall was unaffected (S4B Fig). While weight was unaffected in *Trappc10<sup>-/-</sup>* males, female *Trappc10<sup>-/-</sup>* mice were overweight, showing a rapid weight increase commencing week 10 (S4C Fig). We investigated the cause of the female obesity phenotype and noted a highly significant fat mass increase (P = 9E-08; S4D Fig), alongside a blood cholesterol increase (P = 4E-05) and high-density lipoprotein levels (P = 0.003) (S4E Fig).

We finally sought to further characterize the phenotypical relationship between *Trappc10<sup>-/-</sup>* mice, and a previously published *Trappc9<sup>-/-</sup>* mouse model [30]. Due to the unavailability of live *Trappc9<sup>-/-</sup>* animals, these evaluations were limited to assessments undertaken on historical brain tissue sections. Overall, the impact of TRAPPC10 loss on brain anatomy was more severe than TRAPPC9 loss (S5 Fig). While there were some overlaps between affected brain regions (e.g. corpus callosum reduced size), some neuroanatomical phenotypes were specific to *Trappc10<sup>-/-</sup>* including reduced size of the anterior commissure.

#### Discussion

Here, we define biallelic TRAPPC10 variants as a cause of a neurodevelopmental TRAPPopathy disorder, with core clinical features including microcephaly, severe global developmental delay and intellectual disability, short stature and pervasive behavioural abnormalities (Table 1). Additional variable features include hypotonia, gait abnormalities and seizures. The overlapping phenotypical nature of the TRAPPopathies suggests a common pathomolecular basis of disease involving disruption of TRAPP II/III complex functions. The clinical features of TRAPPC10-related disorder also show extensive overlap with many other Golgipathy disorders [25,26] indicative of a common disease mechanism relating to disrupted Golgi trafficking processes. We observed notable phenotypic overlap with *Trappc10<sup>-/-</sup>* mice which display microcephaly, reduced size of white matter brain structures with hypomyelination and skeletal involvement. Brain abnormalities in mice appear to be more severe than those in humans, in which neuroimaging identified profound thinning of the corpus callosum but no other white matter abnormalities or hypomyelination. However as neuroimaging was only available on a single affected individual who appears to display the least severe phenotype of those assessed, it is possible that more extensive abnormalities may be present in other affected individuals who display a greater degree of microcephaly.

Our genetic studies in combination with our functional investigations on both TRAPPC10<sup>-/-</sup> and patient-derived cells indicate that TRAPPC10 loss of function likely underlies the clinical phenotype. Previous subcellular trafficking studies of TRAPPC10 depleted COS-7 [3] and HEK293 [4] cells determined that TRAPPC10 mediates early Golgi (although not endoplasmic reticulum) trafficking, revealing disrupted Golgi architecture and vesicular accummulation indicative of an anterograde trafficking defect. Our cell model membrane trafficking assays are in complete alignment with such a TRAPPC10 molecular role, defining delayed Golgi trafficking associated with gene KO which importantly could be rescued, though not completely, by WT, and to a lesser extent by mutant TRAPPC10. These findings may in part be explained by differing outcomes of the TRAPPC10 variants, including the possibility that the protein C-terminus plays a specific though as yet unknown post-Golgi secretory pathway role. Interestingly, in previous studies the yeast homologue of TRAPPC10 (Trs130) was rendered conditionally lethal by short truncations in the C-terminal portion of the protein, indicative of a critical functional role of this region [1]. The observed trafficking defects herein may be explained by the loss of GEF activity of TRAPP II towards Rab1 [3], Rab11 [9] and/or Rab18 [4], which alongside other Rab GTPases are key regulators of cellular trafficking by recruiting specific effector proteins to membranes, including lipid droplet homeostasis. Rab18 has specifically been implicated in secretory pathway regulation [31], and RAB18 variants are associated with a neurodevelopmental disorder phenotypically overlapping TRAPPC10-related disorder. Although the specific molecular role of TRAPPC10 in neurological development and function remains unclear, it is tempting to speculate that the predominantly neurological phenoype seen in individuals affected by TRAPPC10-related disorder may be due to the relatively high rates of membrane trafficking shown to be required in the brain for processes such as synapse remodeling [32,33].

An important outcome of our findings was that absence of TRAPPC10 is also associated with a concomitant absence of TRAPPC9. Additionally, although other TRAPP III and core specific components were detected, levels of core TRAPPC2L were reduced suggesting an absence of TRAPP II complex, with no effect on TRAPP III complex. These findings indicate that TRAPPC10 is crucial to TRAPP II complex stability and function. Consistent with our findings, careful examination of the data in Li et al (2017) [4], though not commented on, reveals a decrease in the levels of TRAPPC9 protein in TRAPPC10<sup>-/-</sup> HEK293 cells (see Fig 6D in Ref 4). However, our studies showed TRAPPC10 was still detectable in LCLs lacking TRAPPC9 p.(Arg475<sup>\*</sup>), suggesting that TRAPPC9 may only incorporate into TRAPP II in the presence of TRAPPC10, while the reciprocal is not true. Our overexpression studies of mutant TRAPPC10 provide further insight into this, and in particular identify the expression of full length TRAPPC10 to be of importance for the detection of TRAPPC9 within the TRAPP II complex. Collectively, our results suggest that the absence of the TRAPPC9 (through protein degradation, and consequent loss of TRAPPC9 (through protein degradation or some other mechanism), leading to the absence of TRAPP II.

The concomitant loss of TRAPPC9 associated with pathogenic *TRAPPC10* variants is reflected in the close phenotypical parity of human *TRAPPC9*- (OMIM 613192) and *TRAPPC10*-related conditions (S1 Table) and with both knockout mouse models. Individuals with *TRAPPC9*-related disorder have postnatal-onset microcephaly [34,35]. While the families reported here did not have any recorded birth OFC measurements, there were no reports of affected children having noticably small head size at birth suggesting microcephaly associated with *TRAPPC10*-related disorder may be postnatal. Head growth trajectory was only available for one child, with two measurements available (2.5 and 4.1 years), that indicate microcephaly is present from early childhood and is not progressive. However it remains unclear whether the brain growth was initially normal and then slowed, or whether microcephaly was present

from/before birth. Identification of further affected families will help clarify the nature of the microcephaly in *TRAPPC10*-related disorder, as with *TRAPPC9*-related disorder where this was also unclear initially [19].

Biallelic TRAPPC10 loss of function appears to have a more severe phenotypic impact in mice and humans, compared with biallelic TRAPPC9 loss. In human *TRAPPC10*-related disorder, microcephaly appears to be a relatively consistent feature, (8/10 individuals), as compared to ~60% of TRAPPC9 patients [20,36]. Similarly in mouse models, while the age of onset and progression of microcephaly is unclear, *Trappc10<sup>-/-</sup>* mice display more extensive brain size reduction and loss of white matter structures than *Trappc9<sup>-/-</sup>* mice. These findings may be explained by the concomitant loss of TRAPPC9 observed with TRAPPC10 absence, possibly acting as a double knockout. Obesity is reported in ~50% of individuals with *TRAPPC9*-related disorder and is the only clincial feature not currently observed in individuals with *TRAPPC10*-related disorder. It is important to note that these differences are based on a relatively small sample size, a limitation of this study. Nevertheless, the *Trappc10<sup>-/-</sup>* mice share similar phenotypic outcomes compared to individuals with biallelic *TRAPPC10* variants.

Neuroanatomical studies of *Trappc9*<sup>-/-</sup> mice showed overlapping features with *Trappc10*<sup>-/-</sup> mice including reduced brain size predominantly of white matter structures (S5 Fig) [37], although findings of enlarged striatal size were not identified in *Trappc10*<sup>-/-</sup> mice. TRAPPC9 plays a role in the NF- $\kappa$ B signalling pathway through its interaction with NF- $\kappa$ B inducing kinase (NIK) and the beta subunit of IKK, which both regulate the NF- $\kappa$ B pathway [11,38]. Given the associated loss of TRAPPC9 observed with both the *TRAPPC10* variants and *TRAPPC10*<sup>-/-</sup> cells, it is plausible that TRAPPC10 loss of function may also impact NF- $\kappa$ B signalling pathways through disruption of TRAPPC9, although the involvement of TRAPPC9 in this pathway in brain has been recently questioned [37]. Taken together, we define the genetic, clinical and molecular basis of a novel microcephalic neurodevelopmental disorder associated with biallelic *TRAPPC10* variants in both humans and mice. Our data provides a molecular rationale for the phenotypic overlap between *TRAPPC10*- and *TRAPPC9*-related TRAPPOpa-thy disorders, both involving disruption of TRAPP II-mediated post-Golgi trafficking processes.

#### Material and methods

#### **Ethics statement**

This study was carried out in accordance with institutional ethics review board-approved research protocols from the Institutional Review Board, International Islamic University (IIU), Islamabad, Pakistan. All individuals (or their families) whose data is included in this study provided written informed consent and where applicable specific written consent for publication of photographs.

#### Genetic studies

Genomic DNA was extracted from blood/buccal samples using the ReliaPrep kit (Blood gDNA Miniprep System, Promega) and RNA was extracted using the PAXgene blood miRNA kit and converted to cDNA via reverse transcriptase PCR. Single nucleotide polymorphism (SNP) genotyping was performed using Illumina HumanCytoSNP-12 v2.1 beadchip array, as previously described [39]. WES analysis was performed (NextSeq500; Illumina, San Diego, CA, USA) and involved Agilent Sureselect Whole Exome v6 (Agilent Technologies, Santa Clara, CA) targeting, read alignment (BWA-MEM v0.7.17), mate-pairs fixed and duplicates removed (Picard v2.15), InDel realignment and base quality recalibration (GATK v3.7.0), single nucleotide variant and InDel detection (GATK HaplotypeCaller), variant annotation

(Alamut batch v1.10 and SnpEff (http://snpeff.sourceforge.net/SnpEff\_manual.html))) and read depth assessment (GATK DepthOfCoverage). Copy number variants (CNVs) were detected using SavvyCNV [40]. Variants with <5 reads and/or an allele frequency >0.01 in public genome databases including the Genome Aggregation Database (gnomAD v2.1.1 and v3.1) and the 1000 Genomes Project were excluded. Homozygous and compound heterozygous variants that were exonic and non-synonymous, synonymous with predicted splicing impact or intronic at ±6 nucleotides from splice site were evaluated and cross referenced against the SNP data. Dideoxy sequencing validation of *TRAPPC10* variants was undertaken using standard techniques.

#### Molecular studies

**Construction of TRAPPC10<sup>-/-</sup> cell lines.** Guide RNA (gRNA) design and cloning (CRISPR plasmid pSPCas9(BB)-2A-Puro (PX459) V2.0, Addgene #62988) was performed as described previously [41], with primers: 5'-CACCGCATCTTCGGAGCCCGGCCAT-3' and 5'-AAACATGGCCGGGCTCCGAAGATGC-3'. HEK293 cells were transfected (70% confluency) using JetPRIME with 500ng DNA expressing the gRNA. Following selection (puromycin) and colony expansion, genomic DNA was extracted. The *TRAPPC10* gene region of interest was amplified using oligonucleotide pair: 5'-AGCGTAGTTATGATTTGGGGT-3 and 5'-GCC AAGGAATGAAGGGACAA-3. The resulting PCR product (~540bp, agarose gel electrophoresis) was sequenced. Knockout verification was performed by western analysis (S3 Fig).

**Membrane trafficking assay.** Cells were infected with virus expressing VSVG-GFP-tsO45 (one hour, 37°C). In rescue experiments, cells were transfected with either FLAG-TRAPPC10, FLAG-TRAPPC10 p.Gly1131Valfs\*19 or FLAG-TRAPPC10 p.Pro929Leu constructs, then infected with VSVG–GFP ts045 (one hour, 37°C). Cells were maintained at 40°C overnight. Cycloheximide was added (final concentration, 10µg/ml), prior to shifting cells to 32°C. Time-lapse microscopy started three minutes after temperature shift to allow time to select fluorescent cells (Nikon inverted confocal microscope) as previously described [22]. Movies used for quantitative fluorescence analysis were not subjected to processing. Integrated fluorescence intensity at the Golgi region (defined by the region of perinuclear intensity seen 20–40 minutes after temperature shift) and from whole cell was measured using ImageJ (measurements obtained every 5 minutes). The ratio between fluorescent intensities within the Golgi region and whole cell was generated for each time point. The kinetics of VSVG-GFP-tsO45 trafficking represent change in that ratio over time (0 to 90 min).

**Cell lysis and size exclusion chromatography.** Cells were grown in either DMEM (for HEK293) or RPMI 1640 (for lymphoblastoid cells) medium containing 10% fetal bovine serum (FBS). Cells were lysed in a solution containing 50mM Tris pH 7.2, 150 mM NaCl, 0.5 mM EDTA, 1 mM DTT, 1% Triton X-100 and protease inhibitor cocktail (EDTA-free; Roche). A total of 2–5 mg of protein was fractionated on a Superose 6 Increase 10/300 GL column at a flow rate of 0.4 ml/min. Fractions of 0.5 ml were collected in wash buffer (50 mM Tris pH 7.2, 150 mM NaCl, 0.5 mM EDTA, 1 mM DTT, 0.1% Triton X-100) and probed with the indicated antibodies.

Yeast two hybrid assay. Open reading frames (ORFs) encoding TRAPP proteins were cloned (pGADT7 and pGBKT7 vectors; Clontech). Plasmids were transformed into yeast cells (AH109, Y187). Diploids containing the respective TRAPP ORFs were produced by mating and selecting on synthetic complete medium (SC) lacking leucine and tryptophan. Interactions were assessed on SC medium lacking leucine, tryptophan, histidine and adenine. Serial cell dilutions were spotted onto solid medium. Quantification of interaction was assessed using a  $\beta$ -galactosidase assay employing ONPG and normalized to total protein.

**Molecular biology techniques and reagents.** Standard molecular biological techniques were used to generate FLAG-tagged constructs and the stated *TRAPPC10* variants. Commercially available antibodies used in this study were: anti-TRAPPC2 (rabbit polyclonal, homemade), anti-TRAPPC2L (mouse monoclonal, Santa Cruz sc-377322), anti-TRAPPC3 (rabbit polyclonal, homemade), anti-TRAPPC8 (rabbit polyclonal, Abcam ab122692), anti-TRAPPC9 (rabbit polyclonal, LS Bio LS-C750497), anti-TRAPPC10 (mouse monoclonal, Santa Cruz sc-101259), anti-TRAPPC12 (rabbit polyclonal, homemade), anti-FLAG (mouse monoclonal, Sigma F1804), anti-α-Tubulin (mouse monoclonal, Sigma T6199).

#### Trappc10 knockout mouse model

The mouse model was generated by homologous recombination in embryonic stem cells using the Knockout-first allele method [42], adopting a strategy identifying an exon common to all transcripts (exon 14), upstream of which a LacZ cassette was inserted (S6 Fig). Exon 14 of the *Trappc10* allele, flanked by *loxP* sequences bilaterally, was deleted using a Cre recombinase that recognizes *loxP* sites, producing the *Trappc10*<sup>tmb(EUCOMM)Wtsi</sup> knockout allele. Mice were phenotyped by the Mouse Genetics Project (MGP) pipeline at the Wellcome Sanger Institute, UK (S1 Text).

#### Mouse neuroanatomical studies

The use of mice in the Wellcome Sanger Institute study was carried out in accordance with UK Home Office regulations (license number 80/2076), UK Animals (Scientific Procedures) Act of 1986. Neuroanatomical studies were carried out using three homozygous Trappc10<sup>-/-</sup> and 498 baseline WT mice on a C57BL/6N pure genetic background at 16-weeks of age as previously described [29,43]. We used a recognized statistical model (Gpower) validated for comparison of modest numbers (n = 3) of animals to evaluate neuroanatomical defects with an effect size of 10% or more with 80% detection power [43]. Paraffin embedded brain samples were cut at 5µm thickness (sliding microtome, Leica RM 2145) to obtain coronal brain region at Bregma +0.98 mm and Bregma -1.34 mm according to the Allen Mouse Brain Atlas [44]. Sections were stained with 0.1% Luxol Fast Blue (Solvent Blue 38; Sigma-Aldrich) and 0.1% Cresyl violet acetate (Sigma-Aldrich) and scanned (Nanozoomer 2.0HT, C9600 series) at 20× resolution. Sixty-three brain parameters made up of area and length measurements as well as cell level features, were taken across the two coronal sections (S1 Text). Co-variates, for example sample processing dates and usernames, were collected at every step of the procedure and used to identify data drifts. Using in-house ImageJ plugins, an image analysis pipeline was used to standardize measurements of areas and lengths. Images were quality controlled for the accuracy of sectioning relative to the reference atlas and controlled for asymmetries and histological artefacts. All samples were also systematically assessed for cellular ectopia (misplaced neurons). Data were analyzed using a linear mixed model framework to determine whether a brain region is associated with neuroanatomical defect or not.

#### Supporting information

**S1 Text. Supplementary Material and Methods.** (DOCX)

**S1 Table. Clinical features associated with TRAPPopathy disorders.** Abbreviations: TRAPP; transport protein particles. XLR; X-linked recessive, AR; autosomal recessive, BMI; body mass index, CK; creatine kinase. (DOCX)

**S1 Fig. Clinical features of individuals with biallelic** *TRAPPC10* variants. (A) Individual 1-IV:2 (Family 1) with no evidence of skeletal disproportion. (A-C) Show the facial features of two affected individuals from Family 1 displaying mild craniofacial dysmorphism comprising microcephaly, synophrys and upslanting palpebral fissures. (TIF)

**S2 Fig. Chromatogram of** *TRAPPC10* **c.3392del rtPCR product.** Confirmation of the outcome of the c.3392del; p.(Gly1131Valfs\*19) *TRAPPC10* gene variant on the RNA transcript in blood of an affected individual (individual IV:2, Family 1), demonstrating that this frameshift variant results in a premature stop codon 19 codons downstream. (TIF)

**S3 Fig. Characterization of the** *TRAPPC10<sup>-/-</sup>* **cell lines in HEK293 cells.** Lysate was prepared from parental HEK293 cells and from 3 clones (KO1, KO3 and KO7) that were treated with sgRNA targeting *TRAPPC10*. The lysates were fractionated by SDS-PAGE and probed for TRAPPC10. Tubulin was included as a loading control. (TIF)

S4 Fig. Whole-body phenotype analysis of  $Trappc10^{-/-}$  knockout mouse. (A) Bone length for long bones (Femur and Humerus) in both sexes (n = 7 for all groups) (B) Body length for male  $Trappc10^{-/-}$  (n = 7) and WT (n = 41) and females  $Trappc10^{-/-}$  (n = 7) and WT (n = 27). (C) Body weight curves in grams of male and female  $Trappc10^{-/-}$  mice, between 4 and 16 weeks of age. (D) Fat body composition in grams of male and female  $Trappc10^{-/-}$  mice at 16 weeks of age. (E) *Left*: List of 27 assessed clinical blood chemistry parameters and association in female  $Trappc10^{-/-}$  mice at 16 weeks of age. *Right*: Box plots with raw data points showing results for levels of total cholesterol and high-density lipoprotein (HDL) in millimoles per liter in female  $Trappc10^{-/-}$  mice at 16 weeks of age. Statistical analyses were performed with GraphPad Prism 8.0.2, using two-tailed Student's *t*-tests of equal variances. \*p<0.05 \*\*p<0.01 \*\*\*p<0.001. Arrows indicate directionality of effect and "ns" indicates not significant (p-value>0.05). (TIFF)

**S5 Fig. Neuroanatomical assessment of** *Trappc9* **knockout mouse.** (**A**) Histograms of percentage change relative to  $Trappc9^{+/+}$  (set as 0) for each of the measured parameters. (**B**) Schematic representation of the 22 brain regions quantified at lateral +0.60 mm on sagittal section from  $Trappc9^{+/+}$  (n = 4) and  $Trappc9^{-/-}$  (n = 4) mice. Colored regions indicate the presence of at least one significant parameter within the brain region at the 0.05 level. White indicates a p-value > 0.05, grey shows not enough data to calculate a p-value. (**C**) List of assessed brain parameters.

(TIFF)

S6 Fig. Allelic construction showing targeting of critical exon 14 in the *Trappc10<sup>tm1b</sup>* (EUCOMM)Wtsi mouse.

(TIFF)

**S7 Fig. Representative images for VSV-G trafficking at 0, 20, 40 and 90 minutes.** Wild type HEK293 cells or *TRAPPC10<sup>-/-</sup>* cells either not transfected or transfected with FLAG-tagged wild type TRAPPC10 or one of the *TRAPPC10* variants indicated were infected with VSVG-GFP ts045 4 hours after transfection. After an overnight incubation at 40°C, the cells were shifted to 32°C and imaged every minute. Representative images at 0, 20, 40 and 90 minutes are shown.

(TIF)

#### Acknowledgments

First and foremost, the authors would like to thank the patients and their family members for their participation in this study. We thank members of the Sanger Institute Mouse Pipelines teams and the Research Support Facility for the provision and management of the *Trappc10<sup>-/-</sup>* and *Trappc9<sup>-/-</sup>* mice. We thank Dorinda Wright for the histological work. We are grateful to Dr. John Vincent (Centre for Addiction & Mental Health, Toronto, Canada) for providing the TRAPPC9 p.(Arg475<sup>\*</sup>) LCLs. We would like to thank Professor Christine Hall for reviewing the skeletal radiographs.

#### **Author Contributions**

Conceptualization: Andrew H. Crosby, Michael Sacher, Emma L. Baple.

- Data curation: Lettie E. Rawlins, Hashem Almousa, Shazia Khan, Stephan C. Collins.
- Formal analysis: Lettie E. Rawlins, Hashem Almousa, Shazia Khan, Stephan C. Collins.
- **Funding acquisition:** Lettie E. Rawlins, Binnaz Yalcin, Andrew H. Crosby, Michael Sacher, Emma L. Baple.
- Investigation: Lettie E. Rawlins, Hashem Almousa, Shazia Khan, Stephan C. Collins, Miroslav P. Milev, Joseph Leslie, Djenann Saint-Dic, Valeed Khan, Ana Maria Hincapie, Jacob O. Day, Lucy McGavin, Christine Rowley, Gaurav V. Harlalka, Valerie E. Vancollie, Wasim Ahmad, Christopher J. Lelliott, Asma Gul, Binnaz Yalcin.
- Methodology: Miroslav P. Milev, Binnaz Yalcin, Andrew H. Crosby, Michael Sacher, Emma L. Baple.
- Project administration: Binnaz Yalcin, Andrew H. Crosby, Michael Sacher, Emma L. Baple.
- Supervision: Asma Gul, Binnaz Yalcin, Andrew H. Crosby, Michael Sacher, Emma L. Baple.
- **Visualization:** Lettie E. Rawlins, Hashem Almousa, Shazia Khan, Stephan C. Collins, Andrew H. Crosby, Michael Sacher, Emma L. Baple.
- Writing original draft: Lettie E. Rawlins, Hashem Almousa, Stephan C. Collins, Binnaz Yalcin, Andrew H. Crosby, Michael Sacher, Emma L. Baple.
- Writing review & editing: Lettie E. Rawlins, Hashem Almousa, Stephan C. Collins, Binnaz Yalcin, Andrew H. Crosby, Michael Sacher, Emma L. Baple.

#### References

- Sacher M, Barrowman J, Wang W, Horecka J, Zhang Y, Pypaert M, et al. TRAPP I implicated in the specificity of tethering in ER-to-Golgi transport. Mol Cell. 2001; 7(2):433–42. https://doi.org/10.1016/ s1097-2765(01)00190-3 PMID: 11239471.
- Cai H, Yu S, Menon S, Cai Y, Lazarova D, Fu C, et al. TRAPPI tethers COPII vesicles by binding the coat subunit Sec23. Nature. 2007; 445(7130):941–4. Epub 2007/02/07. https://doi.org/10.1038/ nature05527 PMID: 17287728.
- Yamasaki A, Menon S, Yu S, Barrowman J, Meerloo T, Oorschot V, et al. mTrs130 is a component of a mammalian TRAPPII complex, a Rab1 GEF that binds to COPI-coated vesicles. Mol Biol Cell. 2009; 20 (19):4205–15. Epub 2009/08/05. https://doi.org/10.1091/mbc.e09-05-0387 PMID: 19656848; PubMed Central PMCID: PMC2754934.
- Li C, Luo X, Zhao S, Siu GK, Liang Y, Chan HC, et al. COPI-TRAPPII activates Rab18 and regulates its lipid droplet association. EMBO J. 2017; 36(4):441–57. Epub 2016/12/21. https://doi.org/10.15252/ embj.201694866 PMID: 28003315; PubMed Central PMCID: PMC5694949.
- 5. Westlake CJ, Baye LM, Nachury MV, Wright KJ, Ervin KE, Phu L, et al. Primary cilia membrane assembly is initiated by Rab11 and transport protein particle II (TRAPPII) complex-dependent trafficking of

Rabin8 to the centrosome. Proc Natl Acad Sci U S A. 2011; 108(7):2759–64. Epub 2011/01/27. https://doi.org/10.1073/pnas.1018823108 PMID: 21273506; PubMed Central PMCID: PMC3041065.

- Sacher M, Shahrzad N, Kamel H, Milev MP. TRAPPopathies: An emerging set of disorders linked to variations in the genes encoding transport protein particle (TRAPP)-associated proteins. Traffic. 2019; 20(1):5–26. Epub 2018/09/24. https://doi.org/10.1111/tra.12615 PMID: 30152084.
- Cuenca A, Insinna C, Zhao H, John P, Weiss MA, Lu Q, et al. The C7orf43/TRAPPC14 component links the TRAPPII complex to Rabin8 for preciliary vesicle tethering at the mother centriole during ciliogenesis. J Biol Chem. 2019; 294(42):15418–34. Epub 2019/08/29. https://doi.org/10.1074/jbc.RA119. 008615 PMID: 31467083; PubMed Central PMCID: PMC6802515.
- Riedel F, Galindo A, Muschalik N, Munro S. The two TRAPP complexes of metazoans have distinct roles and act on different Rab GTPases. J Cell Biol. 2018; 217(2):601–17. Epub 2017/12/22. <u>https://doi.org/10.1083/jcb.201705068</u> PMID: 29273580; PubMed Central PMCID: PMC5800803.
- Jenkins ML, Harris NJ, Dalwadi U, Fleming KD, Ziemianowicz DS, Rafiei A, et al. The substrate specificity of the human TRAPPII complex's Rab-guanine nucleotide exchange factor activity. Commun Biol. 2020; 3(1):735. Epub 2020/12/04. https://doi.org/10.1038/s42003-020-01459-2 PMID: 33277614; PubMed Central PMCID: PMC7719173.
- Zong M, Satoh A, Yu MK, Siu KY, Ng WY, Chan HC, et al. TRAPPC9 mediates the interaction between p150 and COPII vesicles at the target membrane. PLoS One. 2012; 7(1):e29995. Epub 2012/01/18. https://doi.org/10.1371/journal.pone.0029995 PMID: 22279557; PubMed Central PMCID: PMC3261171.
- Hu WH, Pendergast JS, Mo XM, Brambilla R, Bracchi-Ricard V, Li F, et al. NIBP, a novel NIK and IKK (beta)-binding protein that enhances NF-(kappa)B activation. J Biol Chem. 2005; 280(32):29233–41. Epub 2005/06/10. <u>https://doi.org/10.1074/jbc.M501670200</u> PMID: <u>15951441</u>; PubMed Central PMCID: PMC3707486.
- Milev MP, Graziano C, Karall D, Kuper WFE, Al-Deri N, Cordelli DM, et al. Bi-allelic mutations in TRAPPC2L result in a neurodevelopmental disorder and have an impact on Rab11 in fibroblasts. J Med Genet. 2018; 55(11):753–64. Epub 2018/08/17. https://doi.org/10.1136/jmedgenet-2018-105441 PMID: 30120216.
- Fiedler J, Le Merrer M, Mortier G, Heuertz S, Faivre L, Brenner RE. X-linked spondyloepiphyseal dysplasia tarda: Novel and recurrent mutations in 13 European families. Hum Mutat. 2004; 24(1):103. https://doi.org/10.1002/humu.9254 PMID: 15221797.
- Al-Deri N, Okur V, Ahimaz P, Milev M, Valivullah Z, Hagen J, et al. A novel homozygous variant in TRAPPC2L results in a neurodevelopmental disorder and disrupts TRAPP complex function. J Med Genet. 2020. Epub 2020/08/25. https://doi.org/10.1136/jmedgenet-2020-107016 PMID: 32843486.
- Van Bergen NJ, Guo Y, Al-Deri N, Lipatova Z, Stanga D, Zhao S, et al. Deficiencies in vesicular transport mediated by TRAPPC4 are associated with severe syndromic intellectual disability. Brain. 2020; 143(1):112–30. <u>https://doi.org/10.1093/brain/awz374</u> PMID: <u>31794024</u>; PubMed Central PMCID: PMC6935753.
- Mohamoud HS, Ahmed S, Jelani M, Alrayes N, Childs K, Vadgama N, et al. A missense mutation in TRAPPC6A leads to build-up of the protein, in patients with a neurodevelopmental syndrome and dysmorphic features. Sci Rep. 2018; 8(1):2053. Epub 2018/02/01. https://doi.org/10.1038/s41598-018-20658-w PMID: 29391579; PubMed Central PMCID: PMC5794855.
- Marin-Valencia I, Novarino G, Johansen A, Rosti B, Issa MY, Musaev D, et al. A homozygous founder mutation in TRAPPC6B associates with a neurodevelopmental disorder characterised by microcephaly, epilepsy and autistic features. J Med Genet. 2018; 55(1):48–54. Epub 2017/06/16. https://doi.org/10. 1136/jmedgenet-2017-104627 PMID: 28626029; PubMed Central PMCID: PMC6056005.
- Mortreux J, Busa T, Germain DP, Nadeau G, Puechberty J, Coubes C, et al. The role of CNVs in the etiology of rare autosomal recessive disorders: the example of TRAPPC9-associated intellectual disability. Eur J Hum Genet. 2018; 26(1):143–8. Epub 2017/11/29. https://doi.org/10.1038/s41431-017-0018-x PMID: 29187737; PubMed Central PMCID: PMC5838970.
- Mir A, Kaufman L, Noor A, Motazacker MM, Jamil T, Azam M, et al. Identification of mutations in TRAPPC9, which encodes the NIK- and IKK-beta-binding protein, in nonsyndromic autosomal-recessive mental retardation. Am J Hum Genet. 2009; 85(6):909–15. https://doi.org/10.1016/j.ajhg.2009.11. 009 PMID: 20004765; PubMed Central PMCID: PMC2790571.
- Abbasi AA, Blaesius K, Hu H, Latif Z, Picker-Minh S, Khan MN, et al. Identification of a novel homozygous TRAPPC9 gene mutation causing non-syndromic intellectual disability, speech disorder, and secondary microcephaly. Am J Med Genet B Neuropsychiatr Genet. 2017; 174(8):839–45. Epub 2017/10/ 14. https://doi.org/10.1002/ajmg.b.32602 PMID: 29031008.
- 21. Liang WC, Zhu W, Mitsuhashi S, Noguchi S, Sacher M, Ogawa M, et al. Congenital muscular dystrophy with fatty liver and infantile-onset cataract caused by TRAPPC11 mutations: broadening of the

phenotype. Skelet Muscle. 2015; 5:29. Epub 2015/08/28. https://doi.org/10.1186/s13395-015-0056-4 PMID: 26322222; PubMed Central PMCID: PMC4551700.

- Koehler K, Milev MP, Prematilake K, Reschke F, Kutzner S, Jühlen R, et al. A novel TRAPPC11 mutation in two Turkish families associated with cerebral atrophy, global retardation, scoliosis, achalasia and alacrima. J Med Genet. 2017; 54(3):176–85. Epub 2016/10/05. <u>https://doi.org/10.1136/jmedgenet-</u> 2016-104108 PMID: 27707803.
- Milev MP, Grout ME, Saint-Dic D, Cheng YH, Glass IA, Hale CJ, et al. Mutations in TRAPPC12 Manifest in Progressive Childhood Encephalopathy and Golgi Dysfunction. Am J Hum Genet. 2017; 101(2):291– 9. https://doi.org/10.1016/j.ajhg.2017.07.006 PMID: <u>28777934</u>; PubMed Central PMCID: PMC5544387.
- Perez Y, Bar-Yaacov R, Kadir R, Wormser O, Shelef I, Birk OS, et al. Mutations in the microtubuleassociated protein MAP11 (C7orf43) cause microcephaly in humans and zebrafish. Brain. 2019; 142 (3):574–85. https://doi.org/10.1093/brain/awz004 PMID: 30715179; PubMed Central PMCID: PMC6391606.
- Passemard S, Perez F, Colin-Lemesre E, Rasika S, Gressens P, El Ghouzzi V. Golgi trafficking defects in postnatal microcephaly: The evidence for "Golgipathies". Prog Neurobiol. 2017; 153:46–63. Epub 2017/04/02. https://doi.org/10.1016/j.pneurobio.2017.03.007 PMID: 28377289.
- Rasika S, Passemard S, Verloes A, Gressens P, El Ghouzzi V. Golgipathies in Neurodevelopment: A New View of Old Defects. Dev Neurosci. 2018; 40(5–6):396–416. Epub 2019/03/15. <u>https://doi.org/10. 1159/000497035</u> PMID: 30878996.
- Santos-Cortez RLP, Khan V, Khan FS, Mughal ZU, Chakchouk I, Lee K, et al. Novel candidate genes and variants underlying autosomal recessive neurodevelopmental disorders with intellectual disability. Hum Genet. 2018; 137(9):735–52. Epub 2018/08/22. https://doi.org/10.1007/s00439-018-1928-6 PMID: 30167849; PubMed Central PMCID: PMC6201268.
- Scales SJ, Pepperkok R, Kreis TE. Visualization of ER-to-Golgi transport in living cells reveals a sequential mode of action for COPII and COPI. Cell. 1997; 90(6):1137–48. <u>https://doi.org/10.1016/s0092-8674(00)80379-7 PMID: 9323141</u>
- 29. Mikhaleva A, Kannan M, Wagner C, Yalcin B. Histomorphological Phenotyping of the Adult Mouse Brain. Curr Protoc Mouse Biol. 2016; 6(3):307–32. Epub 2016/09/01. https://doi.org/10.1002/cpmo.12 PMID: 27584555.
- Liang ZS, Cimino I, Yalcin B, Raghupathy N, Vancollie VE, Ibarra-Soria X, et al. Trappc9 deficiency causes parent-of-origin dependent microcephaly and obesity. PLoS Genet. 2020; 16(9):e1008916. Epub 2020/09/02. https://doi.org/10.1371/journal.pgen.1008916 PMID: 32877400; PubMed Central PMCID: PMC7467316.
- Vazquez-Martinez R, Cruz-Garcia D, Duran-Prado M, Peinado JR, Castaño JP, Malagon MM. Rab18 inhibits secretory activity in neuroendocrine cells by interacting with secretory granules. Traffic. 2007; 8 (7):867–82. Epub 2007/05/04. https://doi.org/10.1111/j.1600-0854.2007.00570.x PMID: 17488286.
- Winkle CC, Gupton SL. Membrane Trafficking in Neuronal Development: Ins and Outs of Neural Connectivity. Int Rev Cell Mol Biol. 2016; 322:247–80. Epub 20160106. https://doi.org/10.1016/bs.ircmb. 2015.10.003 PMID: 26940520; PubMed Central PMCID: PMC4797321.
- Ethell IM, Hagihara K, Miura Y, Irie F, Yamaguchi Y. Synbindin, A novel syndecan-2-binding protein in neuronal dendritic spines. J Cell Biol. 2000; 151(1):53–68. https://doi.org/10.1083/jcb.151.1.53 PMID: 11018053; PubMed Central PMCID: PMC2189810.
- Mochida GH, Mahajnah M, Hill AD, Basel-Vanagaite L, Gleason D, Hill RS, et al. A truncating mutation of TRAPPC9 is associated with autosomal-recessive intellectual disability and postnatal microcephaly. Am J Hum Genet. 2009; 85(6):897–902. https://doi.org/10.1016/j.ajhg.2009.10.027 PMID: 20004763; PubMed Central PMCID: PMC2790576.
- Hnoonual A, Graidist P, Kritsaneepaiboon S, Limprasert P. Novel Compound Heterozygous Mutations in the TRAPPC9 Gene in Two Siblings With Autism and Intellectual Disability. Front Genet. 2019; 10:61. Epub 2019/02/11. https://doi.org/10.3389/fgene.2019.00061 PMID: 30853973; PubMed Central PMCID: PMC6396715.
- Bodnar B, DeGruttola A, Zhu Y, Lin Y, Zhang Y, Mo X, et al. Emerging role of NIK/IKK2-binding protein (NIBP)/trafficking protein particle complex 9 (TRAPPC9) in nervous system diseases. Transl Res. 2020; 224:55–70. Epub 2020/05/17. https://doi.org/10.1016/j.trsl.2020.05.001 PMID: 32434006; PubMed Central PMCID: PMC7442628.
- Ke Y, Weng M, Chhetri G, Usman M, Li Y, Yu Q, et al. Trappc9 deficiency in mice impairs learning and memory by causing imbalance of dopamine D1 and D2 neurons. Sci Adv. 2020; 6(47). Epub 2020/11/ 18. https://doi.org/10.1126/sciadv.abb7781 PMID: 33208359; PubMed Central PMCID: PMC7673810.

- Zhang Y, Liu S, Wang H, Yang W, Li F, Yang F, et al. Elevated NIBP/TRAPPC9 mediates tumorigenesis of cancer cells through NFkB signaling. Oncotarget. 2015; 6(8):6160–78. https://doi.org/10.18632/ oncotarget.3349 PMID: 25704885; PubMed Central PMCID: PMC4467429.
- Rawlins LE, Jones H, Wenger O, Aye M, Fasham J, Harlalka GV, et al. An Amish founder variant consolidates disruption of CEP55 as a cause of hydranencephaly and renal dysplasia. Eur J Hum Genet. 2019; 27(4):657–62. Epub 2019/01/08. https://doi.org/10.1038/s41431-018-0306-0 PMID: 30622327; PubMed Central PMCID: PMC6420058.
- Laver et al. SavvyCNV: Genome-wide CNV calling from off-target reads. PLoS Comput Biol. 2022 Mar 16; 18(3):e1009940. https://doi.org/10.1371/journal.pcbi.1009940
- Ran FA, Hsu PD, Wright J, Agarwala V, Scott DA, Zhang F. Genome engineering using the CRISPR-Cas9 system. Nat Protoc. 2013; 8(11):2281–308. Epub 2013/10/24. <a href="https://doi.org/10.1038/nprot.2013">https://doi.org/10.1038/nprot.2013</a>. 143 PMID: 24157548; PubMed Central PMCID: PMC3969860.
- Skarnes WC, Rosen B, West AP, Koutsourakis M, Bushell W, Iyer V, et al. A conditional knockout resource for the genome-wide study of mouse gene function. Nature. 2011; 474(7351):337–42. Epub 2011/06/15. https://doi.org/10.1038/nature10163 PMID: 21677750; PubMed Central PMCID: PMC3572410.
- Collins SC, Mikhaleva A, Vrcelj K, Vancollie VE, Wagner C, Demeure N, et al. Large-scale neuroanatomical study uncovers 198 gene associations in mouse brain morphogenesis. Nat Commun. 2019; 10 (1):3465. Epub 2019/08/01. https://doi.org/10.1038/s41467-019-11431-2 PMID: 31371714; PubMed Central PMCID: PMC6671969.
- 44. Sunkin SM, Ng L, Lau C, Dolbeare T, Gilbert TL, Thompson CL, et al. Allen Brain Atlas: an integrated spatio-temporal portal for exploring the central nervous system. Nucleic Acids Res. 2013; 41(Database issue):D996–D1008. Epub 2012/11/28. <u>https://doi.org/10.1093/nar/gks1042</u> PMID: <u>23193282</u>; PubMed Central PMCID: PMC3531093.